Author’s response to reviews

Title: The precarious supply of physical therapists across Canada: Exploring national trends in health human resources (1991 to 2005)

Authors:

Michel D Landry (mike.landry@utoronto.ca)
Thomas C Ricketts (ricketts@scher.sr.unc.edu)
Molly C Verrier (m.verrier@utoronto.ca)

Version: 5 Date: 6 September 2007

Author’s response to reviews:

Authors: Michel D Landry, Thomas C Ricketts and Molly C Verrier
Title: The precarious supply of physical therapists across Canada: Exploring national trends in health human resources (1991 to 2005)
Journal: Human Resources for Health
MS: 1416563957142330

Greetings,

As requested, we have completed the following formatting changes:

1) Title page: All authors names were placed on a single line.

2) Affiliations: All city and country details were removed from brackets (actually, the brackets were basically deleted).

3) All wording in text that indicated where figures and tables should be placed were removed.

4) Text was reviewed one more time for accuracy.

Thank you.

MDL

Michel D. Landry PT, PhD
Assistant Professor, Department of Physical Therapy,
University of Toronto, 160-500 University Avenue
Toronto, Ontario, Canada, M5G 1V7
Tel: 416-946-8560
Fax: 416-946-8562
Email: mike.landry@utoronto.ca

President, Canadian Physiotherapy Association and Adjunct Assistant Professor, School of Public Health, University of North Carolina at Chapel Hill (USA)